Author's response to reviews

Title: Functional Analysis of the Involvement of apurinic/apyrimidinic endonuclease 1 in the Resistance to Melphalan in Multiple Myeloma

Authors:

Jiayin Xie (xiejia505@sina.com)
Liang Zhang (zhliang08@hotmail.com)
Mengxia Li (lee800265@hotmail.com)
Jia Du (61169601@qq.com)
Liwei Zhou (351784759@qq.com)
Senlin Yang (vinsonysl@gmail.com)
Linli Zeng (linli.zeng@hotmail.com)
Zengpeng Li (lizengp@hotmail.com)
Ge Wang (wangge70@hotmail.com)
Dong Wang (dongwang64@hotmail.com)

Version: 2 Date: 9 October 2013

Author's response to reviews:

Dear Editor:

We appreciated that you gave us such a good opportunity to revise our manuscript Functional Analysis of the Involvement of apurinic/apyrimidinic endonuclease 1 in the Resistance to Melphalan in Multiple Myeloma (MS#: 1807877949103416). Based on the editor’s and reviewers' comments, we finished some additional experiments, including testing APE1 induction by melphalan in RPMI-8226/LR5 cells and cell viability under E3330 inhibitor pretreatment, to improve our paper in last two months. In addition, we invited a native speaker to edit the English of the manuscript. After a careful revision following the comments, we submit this revision for your review. We have responded point by point to reviewer's comments as listed in the file named Response to Reviewers, along with a clear indication of the location of the main changes. For the main text, we uploaded two versions, one with track change mode on, one is clean text, for your reference.

Thank you very much for your patience.

Best Regards.

Yours Sincerely,

Prof. Jiayin Xie

Address: Cancer Center, Research Institute of Surgery, Daping Hospital, Third Military Medical University, No.10 Changjiang zhilu, Chongqing 400042, China
Tel: 86-23-68757171.
Fax: 86-23-68894062
E-mail: xiejia505@sina.com